**ORIGINAL ARTICLE** 



# Long-term roosting data reveal a unimodular social network in large fission-fusion society of the colony-living Natterer's bat (*Myotis nattereri*)

Veronika M. Zeus<sup>1</sup> · Christine Reusch<sup>1</sup> · Gerald Kerth<sup>1</sup>

Received: 20 December 2017 / Revised: 11 May 2018 / Accepted: 23 May 2018 / Published online: 31 May 2018 © Springer-Verlag GmbH Germany, part of Springer Nature 2018

## Abstract

In many social animals, groups recurrently split into subgroups that regularly re-merge. Such fission-fusion behavior allows individuals to better balance the cost and benefits of group living. However, maintaining a large number of close social links in groups with fission-fusion dynamics may be difficult. It has been suggested that this is the reason why in several species, large groups show more subunits (higher modularity) than do small ones. Many bat species exhibit fission-fusion dynamics in their colonies. This makes them well suited to investigate the proposed link between group size, stability of social links, and group modularity. We studied the daily roosting associations of a Natterer's bat colony (*Myotis nattereri*), where up to 80 members carried individual RFID-tags. Based on more than 10,000 individual recordings, we analyzed the influence of relatedness, age, sex, and breeding status on the colony's social network structure during three breeding seasons. We found an almost fully connected social network with very low modularity and generally weak pairwise associations. Nevertheless, the relative strengths of associations between individuals remained stable across years. Sex, age, and breeding status significantly influenced the strength of an individual's associations and determined the influence of individuals in the network. In general, associations between bats that were similar in all abovementioned traits were stronger than those between dissimilar individuals. Our results show that despite high fission-fusion dynamics, large colony sizes, and low modularity of their social network, Natterer's bats were able to maintain stable long-term associations.

## Significance statement

For a variety of social and ecological reasons, large social groups often consist of several communities with stronger individual bonds within and weaker individual bonds between such social subunits. Unlike that predicted for its relatively large size, the studied Natterer's bat colony that consisted of up to 80 individually marked bats was not subdivided into communities. Despite the fully connected network, the individual associations were not random. Instead, their strength was mainly driven by relatedness and similarity in age and breeding status of the colony members. Moreover, we found stable long-term pairwise relationships between individuals across several years. Our study shows that despite the strong fission-fusion behavior and large size of their colony, Natterer's bats formed a fully connected, unimodal social network.

Keywords Bats · Social network · Association · Assortativity · Fission-fusion · Modularity

Communicated by G. S. Wilkinson

**Electronic supplementary material** The online version of this article (https://doi.org/10.1007/s00265-018-2516-4) contains supplementary material, which is available to authorized users.

Veronika M. Zeus veronika.zeus@gmail.com

<sup>1</sup> Zoological Institute and Museum, Greifswald University, Loitzerstrasse 26, 17489 Greifswald, Germany

# Introduction

In social animals, the strength of individual associations between group members often show a non-random pattern (Krause and Ruxton 2002; Krause et al. 2007). In general, animals may be passively associated with each other in space and time, for example because they share strategies for use of habitat and resources (Mitani et al. 1991; Chaverri et al. 2007; Aplin et al. 2013). Alternatively, they may be actively attracted to—or repelled by—certain other individuals. In this case, their associations reflect individual preferences for each other (Gompper et al. 1997; Krützen et al. 2003; Archie et al. 2006; Mitani 2009). The strengths of individual associations are often influenced by individual preferences or demands that arise from differences in individual traits such as age, kinship, breeding status, or sex (Kerth and König 1999; Wolf et al. 2005; Mitani 2009; Patriquin et al. 2010; Wey and Blumstein 2010). Consequently, such traits often also determine the position and connectivity of individuals in the social network of a group, which reflects individual associations among group members (Wittemyer et al. 2005; Ramos-Fernández et al. 2009; Kerth et al. 2011). In many social networks, individuals show assortative mixing and form subunits with individuals that are similar to themselves with regard to traits such as those mentioned above (Newman 2003).

In order to balance individual needs and group preferences, groups can temporarily split into subgroups that later fuse again, a phenomenon called "fission-fusion" behavior (Aureli et al. 2008). In recent years, various studies investigated the fission-fusion dynamics of social groups in animals (e.g., Lehmann and Boesch 2004; Archie et al. 2006; Popa-Lisseanu et al. 2008; Ramos-Fernández et al. 2009; Kerth et al. 2011). Several of these studies show that despite the high fission-fusion dynamics of their groups, at least some individuals are able to maintain long-term associations (Archie et al. 2006; Patriquin et al. 2010; Kerth et al. 2011). Furthermore, in species with fission-fusion behavior, group size seems to predict group modularity as social links between individuals become more differentiated with increasing numbers of group members (Kudo and Dunbar 2001; Lehmann and Boesch 2004; Kerth et al. 2011; Baigger et al. 2013). It has been discussed that a limited number of close social relationships arising from the modular structure of groups might be a strategy to reduce disadvantages of large groups such as intragroup competition, vulnerability to diseases, or infanticide risk (Smith et al. 2008; Kashima et al. 2013). Alternatively, it could be a consequence of time constraints (Dunbar 1991, 1992a; Lehmann et al. 2007) or cognitive constraints (Dunbar 1992b; Kudo and Dunbar 2001; Wilkinson 2003; Lehmann et al. 2007; Kerth et al. 2011) that limit the number of close relationships that can be maintained in the long term.

Fission-fusion behavior is widespread among bats, but only a minority of species have been studied in enough detail to characterize their colony structure (see reviews by (Kerth 2008; Johnson et al. 2013). For Bechstein's bats, a strong link between the size and the modularity of colonies has been found and it has been suggested that individuals can maintain long-term relationships with only a limited number ( $\leq 20$ ) of colony members (Kerth et al. 2011; Baigger et al. 2013). In the present study, we investigated the colony structure of Natterer's bats (*Myotis nattereri*), a species which is widespread throughout Europe and beyond (Dietz and Kiefer 2014). During summer, Natterer's bats typically roost in tree cavities or bat boxes in forests, although occasionally, buildings are occupied as well (Smith and Racey 2005). Both sexes are highly philopatric (Rivers et al. 2005, 2006; Halczok et al. 2017). Females typically return to their natal colony, where they give birth and raise their young during the maternity season in summer. While some males can be found in these colonies, males have also reported to roost solitary or in all-male groups in the vicinity of their natal colony (Červený and Horáček 1981; Park et al. 1998; Rivers et al. 2006). During autumn, maternity colonies dissolve and the bats migrate to hibernation sites, which are often dozens of kilometers away from the summer habitat (Rivers et al. 2005; Dietz et al. 2007).

As they form colonies of up to 80 individuals, which show a strong fission-fusion behavior (Zeus et al. 2017), Natterer's bats are well suited for a study that aims to assess the social network structure in large and highly dynamic social groups. So far, however, very little is known about the fine scale social structure of Natterer's bats. To our knowledge, previously only two ringing studies addressed this issue (Park et al. 1998; August et al. 2014). The available data suggest that, unlike in many other European bat species, mixed sex groups may be common, with females having a higher number of associates (degree centrality; August et al. 2014). There is also some evidence for the existence of long-lasting but weak individual roosting associations between colony members (Park et al. 1998; August et al. 2014). The caveat of ringing studies is that they depend on capturing the bats from their roosts, and thus do not allow for monitoring roosting associations without disturbing the colony. In an effort to minimize stress on the animals, intervals between data collection in these studies were very long, resulting in very low numbers of recaptures per individual and year (August et al. 2014). Consequently, these studies did not allow for a detailed analysis of the daily fission-fusion behavior of the bats, group modularity, and the stability of social links within colonies over different time periods. Moreover, to our knowledge, no studies on Natterer's bats investigated the influence of kinship, age, and breeding status on the strength of roosting associations among colony members and on their position in the colony's social network.

During three consecutive years, between May and September, we daily monitored 134 bat boxes that were regularly used by one large colony of Natterer's bats as day roosts. In this colony, up to 80 colony members carried individual radio-frequency identification (RFID)-tags. Using automatic RFID-loggers attached to the bats' communal day roosts (bat boxes), we were able to quantify the individual roosting associations without disturbing the bats (Zeus et al. 2017). Based on these individual roost association data, we created social networks and assessed the influence of age, sex, and individual relatedness on the strength of the roosting associations between colony members.

Due to the large size of the colony, we hypothesized that its social network shows a high modularity and comprises several communities, as has been reported for Bechstein's bats that use the same type of roosts (Kerth et al. 2011). We also expected membership within communities to be stable over time with colony members maintaining long-term individual associations (Kerth et al. 2011; Baigger et al. 2013; August et al. 2014). We further assumed that individual traits, namely sex, age class, relatedness, and breeding status, influence the summed strength of an individual's associations (node strength in the colony's network). Finally, we expected assortative mixing between individuals that share the same individual traits and are, therefore, likely to have similar preferences and demands.

## Methods

# **Data collection**

The study took place each May to September in the years 2013 to 2015 in a forest near the city of Würzburg, Germany. The area is home to one large colony of Natterer's bats (*Myotis nattereri* sensu stricto; Puechmaille et al. 2012). Every year before the 20th of May, when the females became visibly pregnant, and after the fledging of the offspring in late July and August, all bats were captured from their day roosts by hand (about two to three times per year). All unmarked individuals, adults and juveniles, were marked with RFID-tags that were inserted via subcutaneous injection, and DNA samples from the wing membrane were taken for subsequent analysis of relatedness (Kerth et al. 2002).

At the same capture events, we assessed the sex, age, and breeding status of the bats: As the colony members were first marked in 2011, it was not possible to determine the exact age of individuals that had already been adult in 2011. Therefore, we defined three distinct age classes: "juvenile" (born in the respective year of capture), "yearlings" (adult bats that had been captured as juveniles in the previous year) and "older bats" (adult bats that are two or more years old). We distinguished between juveniles (only present in July to September) and adult bats (yearlings and older bats) by examining the epiphyseal gap of the fourth metacarpal (Anthony 1988). Juveniles were not included in subsequent analyses because we expected them to still depend strongly on the presence of their lactating mother and thus not to be able to independently choose their day roosts. If bats were specified as adult in the year of their first capture, these individuals were labeled with age "unknown" for that year, as it was not possible to discriminate between yearlings and older without knowing the year of birth. In the following year, such bats were labeled as older

because then they had been at least 2 years old. Adult females were regarded as "breeding" if the bats showed worn patches around the nipples (Racey 1988) or "non-breeding" in the absence of such patches. We classified the breeding status of adult females as "unknown" if they were not captured during that time period or only so late in summer that signs of lactation were no longer explicit. All males that were present in the maternity colonies were classified as non-breeding as males of temperate bat species do not participate in the rearing of the young (Kunz and Hood 2000).

In previous years, 134 bat boxes (type 2FN, Schwegler, Germany) had been installed in the study area (Zeus et al. 2017). These boxes have been regularly used as communal day roosts by the colony during the study period. Each year, data recording started after the last marking event mid of May and stopped in September, when the colony started to dissolve. Roost associations were recorded for a total of 341 days (117 days in 2013, 114 days in 2014, and 110 days in 2015) with a radio-frequency identification (RFID) system that is well established for the automatic monitoring of bats in our study site (Kerth and König 1999; Kerth et al. 2011; Zeus et al. 2017). Every day, all boxes were checked visually without opening them, using an established monitoring protocol (Kerth and Reckardt 2003; Kerth et al. 2011). Occupied boxes were equipped with automatic RFID-tag readers (reader type "LID 650" by EURO I.D.) to monitor the individual group composition in the day roosts. The logger antennae were placed in the boxes' entrances in a way that bats that passed the entrance were recorded by their RFID-tag numbers together with the respective date and time. This set-up has proved to be very reliable in a study by Kerth and Reckardt (2003), where 97% of the bats passing the antenna had been recorded. Bats that were recorded during the emergence from the bat box in the evening were assigned as having spent the day associated with each other in the respective box (Kerth and Reckardt 2003; Kerth et al. 2011). As all association data were recorded automatically, there was little need for blinded methods. The rate of logger malfunctions due to technical problems such as battery failure or antennae damage was very low (3.7% of all monitored day roosts) and the respective days were excluded from the dataset.

## **Genetic analysis**

Following the approach of Halczok et al. (2017), we used 15 nuclear microsatellite markers to assess relatedness between individuals (see supplement for more information). Markers were tested for Hardy-Weinberg equilibrium with Genepop version 4.5.1 (Raymond and Rousset 1995) and for the presence of Null-Alleles with Micro-Checker version 2.2.3 (van Oosterhout et al. 2004) with 10,000 iterations and a 95% confidence interval. In order to confirm natal philopatry in the colony, we identified mother-offspring pairs with Cervus

version 3.0.7 (Kalinowski et al. 2007). The relative relatedness between pairs of individuals was calculated with TrioML (Wang 2007) using Coancestry version 1.0.1.5 (Wang 2011).

#### **Network analysis**

In social network analyses, animals with fewer than a certain number of observations are often excluded because they may cause misinterpretations of the social structure (Whitehead 2008; Farine and Whitehead 2015). The determination of a threshold of minimum number of observations, however, has been handled in various ways in different studies. Many studies have excluded all data below a more or less arbitrarily selected number of observations. In our study, we followed the approach of Aplin et al. (2013) and tested for a relationship between the number of observations and the number of observed association partners. We analyzed each year separately and plotted the number of observations against the number of different association partners (binary degree), which resulted in a saturation curve. We excluded all individuals that did not reach the plateau of the curve. This resulted in the omission of all individuals below 12 observations in 2013 (3 individuals; 5.8% of all analyzed bats in that year), all individuals below 10 observations in 2014 (9 individuals; 11.0%), and all individuals below 20 observations in 2015 (6 individuals; 7.0%). We do not expect any bias in the properties of the network by excluding such a limited number of bats with few observations, as according to Silk et al. (2015), as little as 30% of individuals can be enough to produce a conclusive network.

All network analyses were performed in "R" version 3.3.1 (R Core Team 2016). From our observed roosting data, we generated a weighted, undirected association network by using the R package "asnipe" (Farine 2016a) with values ranging between 0 (never found roosting together) and 1 (always found roosting together). There is a wide variety of different association indices to compute associations between individuals, most of which correct for potential observation biases in the obtained data by giving certain observations a higher or lower weight (Whitehead 2008). However, such weightings are often somewhat arbitrary, generally do not result in a better performance, and can even impede the interpretability of results (Hoppitt and Farine 2017). Therefore, we decided to use the "Simple Ratio Index" (SRI) (Cairns and Schwager 1987; Ginsberg and Young 1992) that needs no such corrections.

Due to the high reliability of the automatic RFID-monitoring, we are highly confident concerning the collected data on the presence or absence of individuals in our monitored boxes at any given day. Nevertheless, we only recorded on average 47% of all colony members per day (minimum 1%, maximum 98%). This can be mainly attributed to two factors: The presence of unmonitored, natural roosting opportunities (tree cavities) at the study site and the departure of individuals at the end of summer. For both reasons, the probability of failing to record two individuals when they are together is the same as failing to record them when they are apart and in such a case, the SRI is unbiased (Hoppitt and Farine 2017).

In order to assess whether the observed network consisted of several subunits (communities), we used the "infomap" algorithm (Rosvall and Bergstrom 2008) as, according to Fortunato and Lancichinetti (2009), it is the best performing algorithm for undirected, weighted networks. Based on this output, we then calculated the modularity of the network. We used the R package "igraph" (Csardi and Nepusz 2006) for both the community detection via "infomap" and the subsequent computation of modularity.

To test whether the pairwise associations in the observed network are non-random, we conducted 100,000 data stream permutations with one random swap per permutation (Farine and Whitehead 2015), as implemented in the "asnipe" package. Subsequently, we computed a new association matrix after every 100 swaps, which resulted in 1000 permuted random networks. In the following, we calculated the proportion of times the coefficients of variation (CVs) of the 1000 permuted random networks were larger than the CV of the observed network (Farine and Whitehead 2015). When 95% of the CVs from the random data were smaller than the CV of the observed data, the test was declared significant at the 5% level. This method was also used for all other analyses that required permutation tests as mentioned below.

In order to test whether the parameters sex, age class, and breeding status influence an individual's node strength, we conducted models of both the observed data and the random networks (Farine and Whitehead 2015). We applied restricted maximum likelihood linear-mixed effect models (LMMs) with normal errors to assess differences in node strength between reproductive classes and between age classes. Since individuals were repeatedly sampled across the years, we included the individual ID as a random intercept in our model as individual association behavior might not be independent between years (see our analyses on temporal stability of associations) despite the complete disintegration of the colony during hibernation. We used a linear model (LM) for the influence of sex on node strength, as this analysis was only possible for 2013 due to the absence of males in the other 2 years. The influence of breeding status and age class was also tested for each year separately via LM and the results thereof can be found in Table S1 in the supplement. For each individual, we calculated the weighted degree to measure node strength which was used as the response variable in the linear models. The attributes sex, age class, and breeding status that were determined during the captures were set as the predictors in our model. P values were computed from the regression coefficients of the observed and random networks and corrected via Holm's Sequential Bonferroni Procedure (Holm 1979; Bretz et al. 2016) in order to avoid Alpha errors due to multiple comparisons. We also conducted the same analysis with eigenvector centrality—a measure of influence (Borgatti 2005)—as the response variable. However, as both network parameters were highly correlated for each year (Kendall rank correlation coefficient (Kendall 1938) 2013: 0.961; 2014: 0.972; 2015: 0.979), the results concerning eigenvector centrality can be found in the supplement (Tables S2 and S3).

To test for a relationship between the association indices and pairwise relatedness, we used the multiple regression quadratic assignment procedure (MRQAP) that is implemented in the "asnipe" package. More specifically, we used the MRQAP with custom permutation networks from the data stream permutations. The association rate was set as the dependent variable and pairwise relatedness as the independent variable.

Moreover, we used the R package "assortnet" (Farine 2016b) to test for assortative mixing of the colony, a weighted measure to assess the extent to which individuals are associated with others that are similar to themselves. Thereby, sex, age, and breeding status were taken into account. The weighted assortativity coefficients of the observed data and the permuted networks were used to compute the p value. Moreover, the output allowed deeper insights into the distribution of association values (edge weights). As there might be a bias for classes that contain proportionally more individuals, we tested via permutation tests if the edge weights between and within classes were higher or lower than expected by chance.

#### **Temporal analysis**

To assess the temporal stability of the observed pairwise associations during the course of one maternity season, we calculated the lagged association rates (LARs) (Whitehead 2008), which is the probability of re-association after a certain period of time. To do so, we used the "asnipe" package and set the step length to 1 day. For this analysis, we used all nonjuvenile individuals of the colony, including those that were previously excluded for the network analysis because of their low number of observations in some years. This was done in order to avoid positively biasing the LARs by only restricting the dataset to frequently observed individuals (Baird and Whitehead 2000; Whitehead 2008). To facilitate the interpretation of the LARs, we further determined the lagged rate of association null, which represents animals associated randomly. This rate was calculated by the mean group size experienced by an individual divided by the mean number of total associates of each individual (mean binary degree) (Farine 2013). We also calculated the lagged identification rate (LIR) which estimates the probability of re-identification after a certain time lag (Whitehead 2001). This measure provides information on demographic effects such as survival or emigration (Whitehead 2001) which might have influenced the LARs. The standard errors of the LARs and LIRs were computed via jackknifing.

Finally, in order to test whether associations between individuals remain stable across years, we recalculated the association matrices from the observed data by only including individuals that were present in both years. These matrices were then compared to each other. To do so, we ran a MRQAP analyses by setting the association network in the later year as the dependent variable, generating 1000 permutation networks thereof via data stream permutation as described above, and using the network of the earlier year as the independent variable. Thereby, we were able to test if pairwise associations between years are interdependent and remain stable.

**Data availability statement** The datasets generated and analyzed during the current study are available from the corresponding author on reasonable request.

# Results

## **Network structure**

During the 341 days of the study, we recorded a total of 10,557 individual roosting associations (3028 in 2013, 3307 in 2014, 4221 in 2015). On average, each individual used between 30 and 32 different roost boxes per year (for yearly average with range see Table 1). Thereby, boxes were switched almost daily (see "mean number of consecutive days/box," Table 1). A plot for each year with the number of different roosts against the number of days a bat was present can be found in the supplement (Fig. S1). Regarding the roost use of two associated individuals, a pair of individuals spent on average between 14 and 20 days together per year (for yearly average with range, see Table 1). Thereby, a pair spent on average 1.4 consecutive days in their most frequented roost (Table 1). Thus, the most preferred roost accounted for roughly 20% of a pair's shared roost use (Table 1). A plot for each year with this proportion against the number of days spent together can be found in the supplement (Fig. S2).

In all 3 years, the colony showed high fission-fusion dynamics as can been seen by the percentage of days on which the colony occupied more than one box (2013: 80.3%; 2014: 61.4%; 2015: 77.2%; Zeus et al. 2017). Nevertheless, in each year, the network—that was based on the individual roosting associations (SRI values)—was almost fully connected (see "Network density," Table 2) with very low modularity and thus consisted of only one community (see Fig. 1 for the year 2015). However, associations were non-random ( $CV_{2013}$ : 62.756, mean random  $CVs_{2013}$ : 52.189,  $p_{2013} < 0.001$ ;  $CV_{2014}$ : 59.627, mean random  $CVs_{2014}$ : 52.896,  $p_{2014} < 0.001$ ;  $CV_{2015}$ : 44.584, mean random  $CVs_{2015}$ : 33.327,  $p_{2015}$ 

Table 1 Average roost use of individual bats and associated pairs per year. The range is given in parentheses

Year	2013	2014	2015
Mean number of different boxes per individual	32.2 (5.0-46.0)	27.5 (7.0–45.0)	29.6 (11.0-43.0)
Mean number of consecutive days per box	1.4 (1.1–1.9)	1.4 (1.0–1.8)	1.4 (1.2–1.7)
Mean number of days a pair spent together	20.1 (1.0-62.0)	14.3 (1.0-54.0)	17.1 (1.0-53.0)
Mean number of days spent together in most frequented roost	1.4 (1.0–12.0)	1.3 (1.0–9.0)	1.3 (1.0–7.0)
% of days spent together in most frequented roost	20.7 (5.6–100.0)	22.2 (4.8-100.0)	17.2 (5.6–100.0)

< 0.001). SRI values indicated that in each year, pairs of individuals shared the same roost only about 20% of the monitoring days on average (see "mean SRI values" Table 2; Figs. 1, S3, and S4 in the Supplementary). Furthermore, individuals only associated with a relatively small proportion of colony members each day (mean per year 32–34%, compare "number of marked individuals" and "mean group size," Table 2).

## **Temporal stability of associations**

Within each year, the LARs started at a value of around 0.5 and further decreased, albeit very gradually, with time. This suggests that in all 3 years, only half of the individual associations persisted after 1 day but remained relatively stable across an extended period of time. In 2015, this time frame is roughly limited to 25 days after which the LAR dropped below the lagged rate of association null, indicating that after this time lag, the probability of re-associations was less likely than expected by random associations. For 2013 and 2014, this was only the case at the end of the respective seasons when the LARs became increasingly unstable. In the same timeframe, the slopes of the LIRs alternate between increasing and decreasing which can be an indication that the bats used the monitored bat boxes less consistently.

The colony members were able to maintain long-term associations across years, as the association networks were significantly positively correlated between successive years (MRQAP, 2013–2014: n = 34 individuals, regression coefficient = 0.423, p < 0.001; 2014–2015: n = 57 individuals, regression coefficient = 0.238, p < 0.001) as well as between 2013 and 2015 (n = 28 individuals, regression coefficient = 0.864, *p* < 0.001).

#### Influence of attributes on node strength

Assessing the influence of sex on node strength was only possible for 2013 due to the absence of males in the other 2 years. The LM Strength~Sex revealed that all six male individuals were significantly weaker associated with their association partners compared to the female individuals and their corresponding association partners (Table 3).

The LMM Strength~Breeding status across all 3 years further suggested that females that reproduced in a given year have higher node strength (Table 3) than non-breeding bats (in 2014 and 2015 only non-breeding females; in 2013: non-breeding females and males). Thus, they seem to have stronger links with their association partners. Conversely, according to the LMM Strength~Age across all 3 years,

 
 Table 2
 Attributes of individuals and network parameters in all 3 years.
"Mean binary degree" is the average number of association partners within the respective year. "Mean group size" is the average number of colony members in a roost. "Network density" is the proportion of all possible links between individuals. "Mean SRI value  $\pm$  CV" signifies the

average association index between pairs of individuals and also stating the coefficients of variation in percent. "Modularity" expresses whether a network is divided into several sub-communities; the value of modularity can either be positive or negative with positive values hinting the presence of community structure

Year	2013	2014	2015
Number of marked individuals	49	73	80
Sex: female/male	43/6	73/0	80/0
Age: older/yearling/n.a.	32/7/10	56/10/7	62/16/2
Breeding status: breeding/non-breeding/n.a.	25/7/17	51/7/15	47/12/21
Mean binary degree	47.6 (44-48)	70.7 (63–72)	79 (78–79)
Mean group size	16.432 (11.5-20.3)	23.267 (16.5-29.2)	27.074 (19.7-36.4)
Network density	99.150	98.136	99.968
Mean SRI value $\pm$ CV	$0.224 \pm 60.447$	$0.199 \pm 58.048$	$0.243 \pm 42.870$
Modularity	-3.108624e-15	3.552714e-15	-4.884981e-15



**Fig. 1** The network structure and strength of pairwise associations in 2015, when 80 individuals had been analyzed: The right graph shows the frequency distribution of the strength of pairwise associations with the red vertical line indicating the mean association strength. The left graph presents the network structure that was produced with Gephi 0.9.2. (Bastian et al. 2009) using a force-directed algorithm where short edges between two nodes represent stronger associations between these two

yearlings had significantly weaker associations than older bats (Table 3).

## Assortative mixing

Even though the colony consisted of only one community, we found significant assortative mixing for all traits that were taken into account, namely sex (significant in all years; 2013 by permutation test, and in 2014 and 2015 by the absence of males), age (significant in 2013 and 2014), and breeding status (significant in 2013 and 2015, Table 4). This shows that associations were significantly stronger between individuals that were similar in the measured traits.

Having a closer look at the discrete attributes in years with significant assortative mixing (Table 4), data of 2013 revealed that, with a mean edge weight of 0.263 (95% CI = 0.254 to 0.271), associations between females accounted for 90% of the sum of all edge weights in the network. This effect was not

individuals. While the network is fully connected, for a better visualization, this graph only shows social links that are above average strength. The breeding status of the individuals is color coded (red = breeding, blue = non-breeding, gray = no information available) and the age of the individuals can be found in the node label ("1y" = yearlings, "old" = older bats). Respective figures for the years 2013 and 2014 can be found in the supplementary

a byproduct of their large class size as the *p* value of the permutation test was significant (p = 0.001). This indicates that females did associate with each other stronger than expected by chance. In contrast, edge weights between males with a mean of 0.204 (95% CI = 0.161 to 0.248, proportion of all edge weights = 1.2%) were not significantly higher than expected by chance (p = 0.137), while those between sexes were significantly lower (proportion of all edge weights = 8.8%, p = 0.001) with an average value of 0.090 (95% CI = 0.083 to 0.097). This suggests that female individuals prefer the presence of their own sex, while males might even avoid females. This evidence for social sexual segregation is also supported by the low proportion of males in the colony in 2013 and their total absence in 2014 and 2015.

For the attributes of age and breeding status, we had to deal with a considerable proportion of individuals with unavailable data ("n.a." in Table 2) ranging between 4 and 24% concerning the age and between 27 and 57% concerning the



Fig. 2 Lagged rates: The solid line is the LAR, the dashed line is the lagged rate of association null, and the dotted line the LIR. Vertical lines signify the standard error

breeding status. Thus, for better comprehension, in the following paragraph, 100% reflects the number of individuals with known status only.

In 2013 and in 2014, the years with significant assortative mixing concerning age class, associations between older bats were significantly stronger than expected from random (2013:

mean edge weight = 0.284, 95% CI = 0.272 to 0.296, proportion of all edge weights = 70.2%, p < 0.001; 2014: mean edge weight = 0.218, 95% CI = 0.212 to 0.223, proportion of all edge weights = 77.1%, p = 0.021). Even though, only small proportions of the total edge weights occurred between yearlings (2013: 3.5%; 2014: 1.3%), these were significantly

**Table 3** Results of the LM (*Strength~Sex*, 2013 only) and the LMMs (across all 3 years) assessing relationships between node strength and attribute sex (basic level: female), breeding status (basic level: non-breeding), and age (basic level: older bats). The estimate ( $coef_{observed}$ ) is given with standard error, degrees of freedom, and *t* value, as well as the

mean estimate of the permutated networks (mean  $coef_{random}$ ), and the computed *p* value from the observed and permuted data (corrected via Holm's sequential Bonferroni procedure). Significant *p* values are shown in italics

Model	coef <sub>observed</sub>	SE	df	t	mean coef <sub>random</sub>	р
Node strength~Sex	-6.684	1.189	47	- 5.623	-6.175	0.001
Node strength~Breeding status	4.075	1.008	53	4.042	3.157	< 0.001
Node strength~Age	-2.635	0.922	81	-2.858	- 1.557	< 0.001

stronger than expected by chance in 2013 (mean edge weight = 0.335, 95% CI = 0.294 to 0.367, p < 0.001). In contrast, individuals of different age classes were significantly weaker associated than expected from the permutated data in both years (2013: mean edge weight = 0.235, 95% CI = 0.220 to 0.251, proportion of all edge weights = 26.3%, p < 0.001; 2014: mean edge weight = 0.160, 95% CI = 0.151 to 0.170, proportion of all edge weights = 20.6%, p < 0.001).

Concerning the breeding status, the largest proportions of the total edge weights were found between breeding females (2013: 81.4%, mean edge weight = 0.307, 95% CI = 0.294 to0.320; 2015: 68.0%, mean edge weight = 0.281, 95% CI = 0.276 to 0.287). These associations were significantly stronger than expected by chance in both 2013 (p < 0.001) and 2015 (p< 0.001). Edge weights were significantly lower than expected from random between bats that did not breed in 2015 (mean edge weight = 0.212, 95% CI = 0.190 to 0.235, proportion of all edge weights = 3.1%, p = 0.009). Breeding and nonbreeding bats also associated significantly weaker than expected from the permutated data in 2013 and 2015 (2013: mean edge weight = 0.099, 95% CI = 0.085 to 0.112, proportion of all edge weights = 15.2%, p < 0.001; 2015: mean edge weight = 0.228, 95% CI = 0.220 to 0.237, proportion of all edge weights = 28.8%, p < 0.001).

#### Influence of relatedness on pairwise associations

Pairwise relatedness was significantly positively correlated with the strength of pairwise association between individuals in all 3 years (pairwise relatedness: regression coefficient<sub>2013</sub> = 0.129, *p* value<sub>2013</sub> < 0.001; regression coefficient<sub>2014</sub> = 0.041, *p* value<sub>2014</sub> = 0.001; regression coefficient<sub>2015</sub> = 0.074, *p* value<sub>2015</sub> < 0.001).

# Discussion

In large social groups, individuals often maintain a few strong individual associations and many weak associations to other members of their group which typically results in a modular network with several communities. Such modular social groups have been described for instance in primates (Lehmann and Boesch 2004) and in bat species such as the Bechstein's bat (Kerth et al. 2011; Baigger et al. 2013). The modularity of large animal groups has been interpreted as a strategy to reduce disadvantages of living in large groups such as increased intra-group competition, infanticide risk, or vulnerability to diseases (Smith et al. 2008; Kashima et al. 2013). Alternatively, it might be a result of time and/or cognitive constrains that may prevent individuals from keeping close bonds with many individuals (Dunbar 1991, 1992a; Kudo and Dunbar 2001; Wilkinson 2003; Lehmann et al. 2007; Kerth et al. 2011) or simply a byproduct of how individuals use space and resources. Interestingly, such group modularity did not occur in the large colony of Natterer's bats (up to 80 adult colony members) during this study, although its roostswitching and fission-fusion behavior was similar to that of Bechstein's bat colonies (Zeus et al. 2017). For comparison, strong modularity occurred in Bechstein's bat colonies that compromised between 30 and 40 individuals while it was absent in small colonies of less than 20 bats (Kerth et al. 2011; Baigger et al. 2013).

One notable factor that determined the strength of an individual's associations to other colony members was their relative relatedness with stronger associations between closely related individuals. Besides in our study species, which is known to show high natal philopatry (Rivers et al. 2005, 2006; Halczok et al. 2017), such a positive influence of kinship on association strength has been described in several other species with fission-fusion behavior, e.g., African elephant (Archie et al. 2006), white-nosed coati (Gompper et al. 1997), common vampire bat (Wilkinson 1984), and Bechstein's bat (Kerth et al. 2003, 2011). In all of the aforementioned species, including Natterer's bats (Rivers et al. 2005), mating takes place outside the natal group. With such a breeding strategy, closely related individuals do not run the risk of inbreeding (Godfrey et al. 2014).

Aside from kinship, we found several individual traits to also influence the strength of an individual's associations (node strength). Although we analyzed each parameter separately, some of them might not be independent of each other as for example breeding individuals in maternity colonies are always female (Kunz and Hood 2000; Rivers et al. 2005), and yearlings might have a lower breeding rate than older bats **Table 4**The assortativity of the network with respect to sex, age, andbreeding status in all 3years. The observed assortativity coefficient isgiven with standard error, as well as the mean assortativity coefficient

of the permutated networks (mean  $coef_{random}$ ) and the computed *p* value from the observed and permuted data. Significant *p* values are shown in italics

	Year	Assortativity coefficient	SE	Mean coef <sub>random</sub>	р
Sex	2013	0.163	0.034	0.123	< 0.001
Age	2013	0.245	0.012	0.219	< 0.001
	2014	0.247	0.008	0.238	0.002
	2015	0.072	0.011	0.067	0.077
Breeding status	2013	0.253	0.007	0.230	< 0.001
	2014	0.298	0.006	0.296	0.126
	2015	0.229	0.005	0.219	< 0.001

(Gaisler 1979). Given that in many bat species, including Natterer's bats, females form maternity colonies with the purpose to give birth and communally rear their young (McCracken and Wilkinson 2000), the observed closer bonds between breeding females may not be surprising. While in most bats of the temperate zone maternity colonies also comprise non-breeding females and-depending on the speciesoccasionally males, there are no reports that these individuals contribute actively in the rearing of the young (Kunz and Hood 2000). As a consequence, breeding and non-breeding colony members are not expected to share the same roosting requirements. For instance, while torpor (reduction of body temperature) is generally seen as beneficial for adult bats to optimize their energy budget at times of low food availability, it can slow down the development of the pups (Hoying and Kunz 1998; Kunz and Hood 2000) and interfere with lactation in breeding females (Wilde et al. 1999). Thus, stronger associations between individuals that have similar traits-the observed assortative mixing-might be due to similar needs and shared preferences for certain roost conditions. Yet, it is surprising and requires further studies to answer why the significant assortative mixing in the colony did not result in a modular network.

Despite the high fission-fusion dynamics in the colony, we found associations between individuals to be relatively stable up to a timeframe of around 80 days (see LARs), which equals roughly the duration of pregnancy and lactation period (Swift 2001). Afterwards, until the bats started to leave the study area in September, individuals were less present in the monitored boxes (data not shown) which can also be seen in the slopes of the LIRs. Bats might have used natural tree cavities more often at this time (compare (Kerth et al. 2001) or temporally may have left the study site for mating at so-called swarming sites (Rivers et al. 2005). Both behaviors could explain the observed breakdown of individual association in late summer.

We also found evidence that individual associations persist across several years. This could be explained with individual social bonds being revived after long periods of separation as the maternity colonies disintegrate in autumn and individuals spend the subsequent hibernation period, which lasts several months, mostly solitary (Beaucournu 1958; Hanák et al. 1962; Gaisler and Hanák 1969). However, most of the parameter that potentially influenced the strength of individual associations within each year, namely relatedness, sex, breeding status, and age class-once the individual is older than one year-did not differ (much) between years. For example, of those individuals with sufficient data, 43 individuals out of 53 (87%) did not differ in their breeding status in two successive years, and 8 out of 9 that were included in the analysis had the same breeding status in 2013 and 2015 (not shown in results). Thus, we cannot fully distinguish whether for Natterer's bats the observed long-term associations are the consequence of long-term individual social bonds-as (Kerth et al. 2011) suggested for Bechstein's bats, a species that also shows fissionfusion behavior (Zeus et al. 2017)-or also reflect assortativity due to shared roosting requirements that remain largely unchanged across several years. Nevertheless, as a pair's most frequented roost only accounted for one quarter of its shared roost use on average, it seems highly unlikely that shared roost preferences alone explain the observed individual roosting associations.

We are aware that the results presented in this paper reflect the social network of one single bat colony. As it has been shown for Bechstein's bats, the social network structure might differ between colonies of the same species (Kerth et al. 2011) or even change in the same colony over time (Baigger et al. 2013). However, in both mentioned studies, group size predicted modularity. Thus, our finding of a non-modular network in such a large colony of Natterer's bats shows that bat species with a very similar roosting and fission-fusion behavior can differ drastically in their network structure, as Bechstein's bats already showed modularity at much smaller colony sizes.

To conclude, while the ecologically similar bat species Bechstein's bat and Natterer's bat both form stable long-term associations with stronger links between closely related individuals (Kerth et al. 2003, 2011), they differ considerably in their network structure: Previous studies have shown—in accordance with our findings-that extensive fission-fusion behavior, as present in our study species (Zeus et al. 2017), seems to come at the expense of close individual bonds that may be too costly to maintain with many individuals (Dunbar 1991, 1992b; Kudo and Dunbar 2001; Lehmann et al. 2007). A counter-measure that has been observed in many other species, including the Bechstein's bat, is the segregation of the group into communities with strong bonds within and weak bonds between these social units (Lehmann and Boesch 2004; Kerth et al. 2011). In contrast, Natterer's bats maintained full connectivity with overall weak associations. At the same time, pairwise associations were significantly stronger between individuals that were closely related or similar in age and breeding status (assortativity). It remains puzzling, why and how Natterer's bats were able to maintain stable long-term individual associations despite large colony sizes, high fission-fusion dynamics, and low modularity of their social network.

Acknowledgments We are deeply grateful to Tanja Halczok and Ina Römer for help and advice in the lab and Dr. Damien Farine for important advice on the social network analysis in R. We are also much obliged to the local forestry and conservation departments for support and to Markus Melber as well as Holger Schmidt for help in the field. Furthermore, we would like to thank Prof. Dr. Gloriana Chaverri, one anonymous referee, and Mel Holdermann for providing very helpful comments on the manuscript.

**Funding** This work was supported by the Deutsche Forschungsgemeinschaft DFG (KE 746/6-1) within the priority program "Ecology and species barriers in emerging viral diseases (SPP 1596)" and the DFG Research training Group "Biological Responses to Novel and Changing Environments" (RTG 2010).

#### Compliance with ethical standards

**Conflict of interest** The authors declare that they have no competing interests.

**Ethical approval** All applicable international, national, and/or institutional guidelines for the care and use of animals were followed. Handling and tagging of the bats were conducted under the permits for species protection (55.1-8642.01-2/00) and animal welfare (55.2-2531.01-47/11 and 55.2-2532-2-20) that had been issued by the government of Lower Franconia.

# References

- Anthony EL (1988) Age determination in bats. In: Kunz TH (ed) Ecological and behavioral methods for the study of bats. Smithonian Institution, Washington, pp 47–58
- Aplin LM, Farine DR, Morand-Ferron J, Cole EF, Cockburn A, Sheldon BC (2013) Individual personalities predict social behaviour in wild networks of great tits (*Parus major*). Ecol Lett 16:1365–1372. https://doi.org/10.1111/ele.12181
- Archie EA, Moss CJ, Alberts SC (2006) The ties that bind: genetic relatedness predicts the fission and fusion of social groups in wild African elephants. Proc R Soc Lond B 273:513–522. https://doi. org/10.1098/rspb.2005.3361

- August TA, Nunn MA, Fensome AG, Linton DM, Mathews F (2014) Sympatric woodland *Myotis* bats form tight-knit social groups with exclusive roost home ranges. PLoS One 9:e112225. https://doi.org/ 10.1371/journal.pone.0112225
- Aureli F, Schaffner CM, Boesch C, Bearder SK, Call J, Chapman CA, Connor R, Fiore AD, Dunbar RIM, Henzi SP, Holekamp K, Korstjens AH, Layton R, Lee P, Lehmann J, Manson JH, Ramos-Fernandez G, Strier KB, Schaik CP (2008) Fission-fusion dynamics: new research frameworks. Curr Anthropol 49:627–654. https://doi. org/10.1086/586708
- Baigger A, Perony N, Reuter M, Leinert V, Melber M, Grünberger S, Fleischmann D, Kerth G (2013) Bechstein's bats maintain individual social links despite a complete reorganisation of their colony structure. Naturwissenschaften 100:895–898. https://doi.org/10. 1007/s00114-013-1090-x
- Baird RW, Whitehead H (2000) Social organization of mammal-eating killer whales: group stability and dispersal patterns. Can J Zool 78: 2096–2105. https://doi.org/10.1139/cjz-78-12-2096
- Bastian M, Heymann S, Jacomy M (2009) Gephi: an open source software for exploring and manipulating networks. International AAAI Conference on Weblogs and Social Media, https://gephi. org/
- Beaucournu J-C (1958) Contribution à l'inventaire faunistique des cavités souteraines de l'ouest de la France: 1. Mammiferes. Bull Soc Sc Natur Ouest France 54:5–16
- Borgatti SP (2005) Centrality and network flow. Soc Netw 27:55-71
- Bretz F, Hothorn T, Westfall P (2016) Multiple comparisons using R. CRC Press, Boca Raton, FL
- Cairns SJ, Schwager SJ (1987) A comparison of association indices. Anim Behav 35:1454–1469. https://doi.org/10.1016/S0003-3472(87)80018-0
- Červený J, Horáček I (1981) Comments on the life history of *Myotis* nattereri in Czechoslovakia. Myotis 18:156–162
- Chaverri G, Gamba-Rios M, Kunz TH (2007) Range overlap and association patterns in the tent-making bat Artibeus watsoni. Anim Behav 73:157–164. https://doi.org/10.1016/j.anbehav.2006.06.003
- Core Team R (2016) R: a language and environment for statistical computing. R Foundation for Statistical Computing, Vienna, Austria http://www.R-project.org
- Csardi G, Nepusz T (2006) The igraph software package for complex network research. InterJournal Complex Systems:1695
- Dietz C, Kiefer A (2014) Die Fledermäuse Europas: kennen, bestimmen, schützen. Kosmos, Stuttgart
- Dietz C, von Helversen O, Nill D (eds) (2007) Handbuch der Fledermäuse Europas und Nordwestafrikas: Biologie, Kennzeichen, Gefährdung. Kosmos, Stuttgart
- Dunbar RIM (1991) Functional significance of social grooming in primates. Folia Primatol 57:121–131. https://doi.org/10.1159/ 000156574
- Dunbar RIM (1992a) Time: a hidden constraint on the behavioural ecology of baboons. Behav Ecol Sociobiol 31:35–49
- Dunbar RIM (1992b) Neocortex size as a constraint on group size in primates. J Hum Evol 22:469–493. https://doi.org/10.1016/0047-2484(92)90081-J
- Farine DR (2013) Animal social network inference and permutations for ecologists in R using asnipe. Methods Ecol Evol 4:1187–1194. https://doi.org/10.1111/2041-210X.12121
- Farine DR (2016a) Asnipe: animal social network inference and permutations for ecologists, https://CRAN.R-project.org/package=asnipe
- Farine DR (2016b) Assortnet: calculate the assortativity coefficient of weighted and binary networks, https://CRAN.R-project.org/package=assortnet
- Farine DR, Whitehead H (2015) Constructing, conducting and interpreting animal social network analysis. J Anim Ecol 84:1144– 1163. https://doi.org/10.1111/1365-2656.12418

- Fortunato S, Lancichinetti A (2009) Community detection algorithms: a comparative analysis. In: proceedings of the fourth international ICST conference on performance evaluation methodologies and tools. ICST (Institute for Computer Sciences, Social-Informatics and Telecommunications Engineering), Gent, Belgium, p 27
- Gaisler J (1979) Ecology of bats. In: Stoddart DM (ed) Ecology of small mammals. Springer, Berlin, pp 281–342
- Gaisler J, Hanák V (1969) Summary of the results of bat-banding in Czechoslovakia, 1948-1967. Lynx 10:25–34
- Ginsberg JR, Young TP (1992) Measuring association between individuals or groups in behavioural studies. Anim Behav 44:377–379. https://doi.org/10.1016/0003-3472(92)90042-8
- Godfrey SS, Ansari TH, Gardner MG, Farine DR, Bull CM (2014) A contact-based social network of lizards is defined by low genetic relatedness among strongly connected individuals. Anim Behav 97:35–43. https://doi.org/10.1016/j.anbehav.2014.08.019
- Gompper ME, Gittleman JL, Wayne RK (1997) Genetic relatedness, coalitions and social behaviour of white-nosed coatis, *Nasua narica*. Anim Behav 53:781–797. https://doi.org/10.1006/anbe.1996.0344
- Halczok TK, Fischer K, Gierke R, Zeus VM, Meier F, Treß C, Balkema-Buschmann A, Puechmaille SJ, Kerth G (2017) Evidence for genetic variation in Natterer's bats (*Myotis nattereri*) across three regions in Germany but no evidence for co-variation with their associated astroviruses. BMC Evol Biol 17:5. https://doi.org/10.1186/s12862-016-0856-0
- Hanák V, Gaisler J, Figala J (1962) Results of bat-banding in Czechoslovakia, 1948–1960. Acta Univ Carol–Biol 1:9–87
- Holm S (1979) A simple sequentially rejective multiple test procedure. Scand J Stat 6:65–70
- Hoppitt W, Farine D (2017) Association indices for quantifying social relationships: how to deal with missing observations of individuals or groups. bioRxiv 117044. doi: https://doi.org/10.1101/117044
- Hoying KM, Kunz TH (1998) Variation in size at birth and post-natal growth in the insectivorous bat *Pipistrellus subflavus* (Chiroptera: *Vespertilionidae*). J Zool 245:15–27. https://doi.org/10.1111/j.1469-7998.1998.tb00067.x
- Johnson JS, Kropczynski JN, Lacki MJ (2013) Social network analysis and the study of sociality in bats. Acta Chiropterol 15:1–17. https:// doi.org/10.3161/150811013X667821
- Kalinowski ST, Taper ML, Marshall TC (2007) Revising how the computer program CERVUS accommodates genotyping error increases success in paternity assignment. Mol Ecol 16:1099–1106. https:// doi.org/10.1111/j.1365-294X.2007.03089.x
- Kashima K, Ohtsuki H, Satake A (2013) Fission-fusion bat behavior as a strategy for balancing the conflicting needs of maximizing information accuracy and minimizing infection risk. J Theor Biol 318:101– 109. https://doi.org/10.1016/j.jtbi.2012.10.034
- Kendall MG (1938) A new measure of rank correlation. Biometrika 30: 81–93. https://doi.org/10.1093/biomet/30.1-2.81
- Kerth G (2008) Causes and consequences of sociality in bats. BioScience 58:737–746. https://doi.org/10.1641/B580810
- Kerth G, König B (1999) Fission, fusion and nonrandom associations in female Bechstein's bats (*Myotis bechsteinii*). Behaviour 136:1187– 1202. https://doi.org/10.1163/156853999501711
- Kerth G, Reckardt K (2003) Information transfer about roosts in female Bechstein's bats: an experimental field study. Proc R Soc Lond B 270:511–515. https://doi.org/10.1098/rspb.2002.2267
- Kerth G, Weissmann K, König B (2001) Day roost selection in female Bechstein's bats (*Myotis bechsteinii*): a field experiment to determine the influence of roost temperature. Oecologia 126:1–9. https://doi.org/10.1007/s004420000489
- Kerth G, Safi K, König B (2002) Mean colony relatedness is a poor predictor of colony structure and female philopatry in the communally breeding Bechstein's bat (*Myotis bechsteinii*). Behav Ecol Sociobiol 52:203–210. https://doi.org/10.1007/s00265-002-0499-6

- Kerth G, Almasi B, Ribi N, Thiel D, Lüpold S (2003) Social interactions among wild female Bechstein's bats (*Myotis bechsteinii*) living in a maternity colony. Acta Ethol 5:107–114. https://doi.org/10.1007/ s10211-003-0075-8
- Kerth G, Perony N, Schweitzer F (2011) Bats are able to maintain longterm social relationships despite the high fission-fusion dynamics of their groups. Proc R Soc Lond B 278:2761–2767. https://doi.org/10. 1098/rspb.2010.2718
- Krause J, Ruxton GD (eds) (2002) Living in groups. Oxford University Press, Oxford
- Krause J, Croft DP, James R (2007) Social network theory in the behavioural sciences: potential applications. Behav Ecol Sociobiol 62:15– 27. https://doi.org/10.1007/s00265-007-0445-8
- Krützen M, Sherwin WB, Connor RC, Barré LM, van der Casteele T, Mann J, Brooks R (2003) Contrasting relatedness patterns in bottlenose dolphins (*Tursiops* sp.) with different alliance strategies. Proc R Soc Lond B 270:497–502
- Kudo H, Dunbar RIM (2001) Neocortex size and social network size in primates. Anim Behav 62:711–722
- Kunz TH, Hood WR (2000) Parental care and postnatal growth in the Chiroptera. In: Crichton EG, Krutzsch PH (eds) Reproductive biology of bats. Academic Press, London, pp 415–468
- Lehmann J, Boesch C (2004) To fission or to fusion: effects of community size on wild chimpanzee (*Pan troglodytes verus*) social organisation. Behav Ecol Sociobiol 56:207–216. https://doi.org/10.1007/ s00265-004-0781-x
- Lehmann J, Korstjens AH, Dunbar RIM (2007) Group size, grooming and social cohesion in primates. Anim Behav 74:1617–1629
- McCracken GF, Wilkinson GS (2000) Bat mating systems. In: Crichton EG, Krutszch PH (eds) Reproductive biology of bats. Academic Press, London, pp 321–362
- Mitani JC (2009) Male chimpanzees form enduring and equitable social bonds. Anim Behav 77:633–640. https://doi.org/10.1016/j.anbehav. 2008.11.021
- Mitani JC, Grether GF, Rodman PS, Priatna D (1991) Association among wild orang-utans: sociality, passive aggregations or chance? Anim Behav 42:33–46
- Newman ME (2003) Mixing patterns in networks. Phys Rev E 67:026126
- Park KJ, Masters E, Altringham JD (1998) Social structure of three sympatric bat species (Vespertilionidae). J Zool 244:379–389
- Patriquin KJ, Leonard ML, Broders HG, Garroway CJ (2010) Do social networks of female northern long-eared bats vary with reproductive period and age? Behav Ecol Sociobiol 64:899–913. https://doi.org/ 10.1007/s00265-010-0905-4
- Popa-Lisseanu AG, Bontadina F, Mora O, Ibáñez C (2008) Highly structured fission–fusion societies in an aerial-hawking, carnivorous bat. Anim Behav 75:471–482. https://doi.org/10.1016/j.anbehav.2007. 05.011
- Puechmaille SJ, Allegrini B, Boston ESM, Dubourg-Savage MJ, Evin A, Knochel A, le Bris Y, Lecoq V, Lemaire M, Rist D, Teeling EC (2012) Genetic analyses reveal further cryptic lineages within the *Myotis nattereri* species complex. Mamm Biol 77:224–228. https:// doi.org/10.1016/j.mambio.2011.11.004
- Racey PA (1988) Reproductive assessment in bats. In: Kunz TH (ed) Ecological and behavioral methods for the study of bats. Smithonian Institution, Washington, pp 31–45
- Ramos-Fernández G, Boyer D, Aureli F, Vick LG (2009) Association networks in spider monkeys (*Ateles geoffroyi*). Behav Ecol Sociobiol 63:999–1013. https://doi.org/10.1007/s00265-009-0719-4
- Raymond M, Rousset F (1995) GENEPOP, version 1.2: population genetics software for exact tests and ecumenicism. J Hered 86:248– 249. https://doi.org/10.1093/oxfordjournals.jhered.a111573
- Rivers NM, Butlin RK, Altringham JD (2005) Genetic population structure of Natterer's bats explained by mating at swarming sites and

philopatry. Mol Ecol 14:4299–4312. https://doi.org/10.1111/j.1365-294X.2005.02748.x

- Rivers NM, Butlin RK, Altringham JD (2006) Autumn swarming behaviour of Natterer's bats in the UK: population size, catchment area and dispersal. Biol Conserv 127:215–226. https://doi.org/10.1016/j. biocon.2005.08.010
- Rosvall M, Bergstrom CT (2008) Maps of random walks on complex networks reveal community structure. P Natl Acad Sci USA 105: 1118–1123. https://doi.org/10.1073/pnas.0706851105
- Silk MJ, Jackson AL, Croft DP, Colhoun K, Bearhop S (2015) The consequences of unidentifiable individuals for the analysis of an animal social network. Anim Behav 104:1–11. https://doi.org/10. 1016/j.anbehav.2015.03.005
- Smith PG, Racey PA (2005) The itinerant natterer: physical and thermal characteristics of summer roosts of *Myotis nattereri* (Mammalia: Chiroptera). J Zool 266:171–180. https://doi.org/10.1017/ S0952836905006758
- Smith JE, Kolowski JM, Graham KE, Dawes SE, Holekamp KE (2008) Social and ecological determinants of fission–fusion dynamics in the spotted hyaena. Anim Behav 76:619–636. https://doi.org/10.1016/j. anbehav.2008.05.001
- Swift S (2001) Growth rate and development in infant Natterer's bats (*Myotis nattereri*) reared in a flight room. Acta Chiropterol 3:217–223
- van Oosterhout C, Hutchinson WF, Wills DPM, Shipley P (2004) Microchecker: software for identifying and correcting genotyping errors in microsatellite data. Mol Ecol Notes 4:535–538. https://doi.org/10. 1111/j.1471-8286.2004.00684.x
- Wang J (2007) Triadic IBD coefficients and applications to estimating pairwise relatedness. Genet Res 89:135–153. https://doi.org/10. 1017/S0016672307008798
- Wang J (2011) Coancestry: a program for simulating, estimating and analysing relatedness and inbreeding coefficients. Mol Ecol

Resour 11:141–145. https://doi.org/10.1111/j.1755-0998.2010. 02885.x

- Wey TW, Blumstein DT (2010) Social cohesion in yellow-bellied marmots is established through age and kin structuring. Anim Behav 79: 1343–1352. https://doi.org/10.1016/j.anbehav.2010.03.008
- Whitehead H (2001) Analysis of animal movement using opportunistic individual identifications: application to sperm whales. Ecology 82: 1417–1432
- Whitehead H (2008) Analyzing animal societies: quantitative methods for vertebrate social analysis. University of Chicago Press, Chicago
- Wilde CJ, Knight CH, Racey PA (1999) Influence of torpor on milk protein composition and secretion in lactating bats. J Exp Zool 284:35–41. https://doi.org/10.1002/(SICI)1097-010X(19990615) 284:1<35::AID-JEZ6>3.0.CO;2-Z
- Wilkinson GS (1984) Reciprocal food sharing in the vampire bat. Nature 308:181–184. https://doi.org/10.1038/308181a0
- Wilkinson GS (2003) Social and vocal complexity in bats. In: de Waal FBM, Tyack PL (eds) Animal social complexity: intelligence, culture, and individualized societies. Harvard University Press, Cambridge, MA, pp 322–341
- Wittemyer G, Douglas-Hamilton I, Getz WM (2005) The socioecology of elephants: analysis of the processes creating multitiered social structures. Anim Behav 69:1357–1371. https://doi.org/10.1016/j. anbehav.2004.08.018
- Wolf JBW, Kauermann G, Trillmich F (2005) Males in the shade: habitat use and sexual segregation in the Galápagos sea lion (*Zalophus californianus wollebaeki*). Behav Ecol Sociobiol 59:293–302. https://doi.org/10.1007/s00265-005-0042-7
- Zeus VM, Puechmaille SJ, Kerth G (2017) Conspecific and heterospecific social groups affect each other's resource use: a study on roost sharing among bat colonies. Anim Behav 123:329–338. https://doi.org/10.1016/j.anbehav.2016.11.015